

# **Geophysical Research Letters**



10.1029/2021GL092773

#### **Key Points:**

- Statistical uncertainties in proxy reconstructions are dominated by unexplained variance in the calibration
- Proxy data should be presented as explicitly distinct from environmental interpretations
- Future efforts should prioritize collection of multiple explanatory variables for calibrations, and true replicate paleoanalyses

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

H. L. O. McClelland, hmcclelland@unimelb.edu.au

#### **Citation:**

McClelland, H. L. O., Halevy, I., Wolf-Gladrow, D. A., Evans, D., & Bradley, A. S. (2021). Statistical uncertainty in paleoclimate proxy reconstructions. *Geophysical Research Letters*, 48, e2021GL092773. https://doi. org/10.1029/2021GL092773

Received 2 FEB 2021 Accepted 29 JUN 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Statistical Uncertainty in Paleoclimate Proxy Reconstructions

H. L. O. McClelland<sup>1,2,3</sup>, I. Halevy<sup>3</sup>, D. A. Wolf-Gladrow<sup>4</sup>, D. Evans<sup>5</sup>, and A. S. Bradley<sup>2</sup>

<sup>1</sup>School of Geography, Earth and Atmospheric Sciences, University of Melbourne, Parkville, VIC, Australia, <sup>2</sup>Department of Earth & Planetary Sciences, Washington University in St. Louis, St Louis, MO, USA, <sup>3</sup>Department of Earth & Planetary Sciences, Weizmann Institute of Science, Rehovot, Israel, <sup>4</sup>Alfred-Wegener-Institut Helmholtz Zentrum für Polar- und Meeresforschung, Bremerhaven, Germany, <sup>5</sup>Institute of Geosciences, Goethe University Frankfurt, Frankfurt, Germany

**Abstract** A quantitative analysis of any environment older than the instrumental record relies on proxies. Uncertainties associated with proxy reconstructions are often underestimated, which can lead to artificial conflict between different proxies, and between data and models. In this paper, using ordinary least squares linear regression as a common example, we describe a simple, robust and generalizable method for quantifying uncertainty in proxy reconstructions. We highlight the primary controls on the magnitude of uncertainty, and compare this simple estimate to equivalent estimates from Bayesian, nonparametric and fiducial statistical frameworks. We discuss when it may be possible to reduce uncertainties, and conclude that the unexplained variance in the calibration must always feature in the uncertainty in the reconstruction. This directs future research toward explaining as much of the variance in the calibration data as possible. We also advocate for a "data-forward" approach, that clearly decouples the presentation of proxy data from plausible environmental inferences.

**Plain Language Summary** Earth's surface environments have varied significantly throughout geologic time. Accurate quantification of these ancient environmental changes relies on proxies—materials that are known to change composition or morphology with the ambient environment. These approaches have provided insight into important questions across the Earth sciences, from the context and consequences of biological evolution and volcanic eruptions, to threshold behavior and long-term feedbacks within the modern climate system. Although the uncertainty associated with an estimate of an environmental change is of equal importance to the estimate itself, uncertainties are widely either underestimated or else rely on proxy-specific statistical models. Fortunately, a very good, and broadly applicable, estimate of uncertainty is extremely simple to calculate. In this paper we show that statistical uncertainty in proxy reconstructions is mostly due to the magnitude of scatter around the calibration line. We furthermore attempt to give the reader an intuition for how large reported uncertainties in proxy reconstructions should be, how and when they can be reduced, and where future efforts to reduce uncertainty should be directed.

## 1. Introduction

Ancient environments cannot be observed directly, so paleo-environmental reconstructions rely on indirect proxy approaches. A proxy is a characteristic of a material that can be measured, and is known to correlate with some aspect of the material's environment of formation. A paleo-environmental proxy reconstruction is complete only with a quantitative statement of its uncertainty (Taylor & Kuyatt, 1994). A number of proxy-specific statistical models now exist for a handful of mature proxies, however, statistical uncertainty in proxy reconstructions remains widely underestimated, particularly for new proxies. Our aim is to explore the factors that control the magnitude of uncertainty in reconstructions, and to provide a simple, robust and generalizable approach that can be used to estimate the uncertainty associated with any paleo-environmental proxy reconstruction.

The first step in proxy reconstruction is to constrain a relationship between the proxy variable to be measured (P) and the environmental variable to be inferred (E) with a calibration data set consisting of paired values. For all proxies, the value of P depends on the value of E, so this relationship must be defined with a



forward model, where *E* is the independent variable, and *P* is the dependent variable. Calibration data can be generated in laboratory experiments by varying *E* independently from all other influential variables (e.g., Bemis et al., 1998; Kluge et al., 2015), or by analyzing natural environmental samples, taking advantage of natural temporal or spatial variation in *E* (e.g., Anand et al., 2003; Dekens et al., 2002; Gray et al., 2018; Kim et al., 2008). The second step is to estimate the value of *E* that led to the value of *P*, which involves "inverting" the calibration model. As *P* is always a function of multiple environmental variables in addition to *E*, there is always residual variance in the calibration data, which reflects the complex variability of the natural world. It is this residual variance that usually dominates the uncertainty in proxy reconstructions, yet this component of uncertainty is often ignored.

In our analysis, we focus on the most common type of linear relationship between E and P where the three prerequisites for ordinary least squares (OLS) linear regression are satisfied. Despite its simplicity, OLS linear regression highlights the major problems common to all proxy approaches, and enables us to explore the controls on the magnitude of uncertainty in a reconstruction. A number of widely used proxies are based on linear calibrations, and any nonlinear, multivariate and more complex models can also be coerced into a linear form by algebraic manipulation or by plotting the model-predicted values against the observations.

# 2. Calibration

A calibration data set consists of paired values of *E* and *P* (i.e.,  $(E_i, P_i)$ , i = 1, ..., n). In the calibration context, *E* is considered to be an independent non-stochastic variable. The value of *P* is influenced by *E*, in addition to other, often unknown, factors that are treated in the current approach as noise. *P* is, therefore, a dependent stochastic variable. By convention, the independent variable (*E*) is plotted on the *x*-axis, and the dependent variable (*P*) is plotted on the *y*-axis.

In our example we consider a calibration data set that conforms to the assumptions of the Gauss-Markov theorem. This requires that the noise: (a) has zero mean; (b) has a single variance (is homoscedastic); (c) comprises uncorrelated values. The intercept ( $\beta_0$ ) and the slope ( $\beta_1$ ) of the straight line model are estimated by OLS linear regression, where the line minimizes the sum of squared differences between model *P* values and observations of *P*. This is a "*P*-on-E" OLS linear regression. The data pairs ( $E_i$ ,  $P_i$ ; i = 1, ..., n) are then related by the forward model:

$$P_i = \beta_1 E_i + \beta_0 + \epsilon_i, \tag{1}$$

where  $\epsilon$  is a noise term. We here assume that the noise is normal ( $\epsilon_i \sim \mathcal{N}(0, \sigma^2)$ ), but this is not a requirement for OLS regression. When one of the prerequisites for Equation 1 is not satisfied, an alternative method of regression is required. Nonlinear and multivariate models will not be discussed here, nor will univariate linear models that account for uncertainties in both variables, but these are thoroughly reviewed elsewhere (e.g., Smith, 2009).

This regression model is defined by three parameters:  $\beta_0$ ,  $\beta_1$  and  $\sigma$ . Estimates of these parameters are respectively denoted:  $\hat{\beta}_0$ ,  $\hat{\beta}_1$  and  $\hat{\sigma}$ . Classically, uncertainties in estimates of  $\beta_0$  and  $\beta_1$  are quantified and illustrated by construction of confidence bands (CBs; Figure 1) valid over all *E* values (hyperbolic CB [Scheffé, 1959]) or over restricted intervals (straight lines [Gafarian, 1964]). CBs are constructed in such a way that, under repeated experiments, the long-run average of CBs that cover the true line is the nominal probability, say, 95% (see Casella & Berger, 2002 for a detailed discussion). In the Bayesian approach, all model parameters are estimated by a posterior distribution that integrates information from the data and the model assumptions. Uncertainties in the parameter values decrease with increasing *n*, typically according to  $1/\sqrt{n}$ . Parameter uncertainties can, therefore, be conceptualized as analogous to standard errors, where larger calibration data sets result in greater confidence in the model. However, while more calibration data improves estimates of the intercept, slope and magnitude of the noise, the same is not true for the uncertainty associated with inverse prediction.

## 3. Inverse Prediction

#### 3.1. Inversion of the Forward Model

The goal of a paleo-environmental proxy is to estimate *E* from *P*. For a typical linear relationship where *E* is the independent variable; this is called *inverse prediction*. The classical estimator for  $E(\hat{E}_0; \text{ e.g., Osborne, 1991})$ , can be straightforwardly derived from Equation 1 as follows:

$$\hat{E}_0 = \frac{P_0 - \hat{\beta}_0}{\hat{\beta}_1},$$
(2)

where  $P_0$  is a single new value of the proxy variable, and where  $\hat{\beta}_0$  and  $\hat{\beta}_1$  are respectively the estimates of  $\beta_0$  and  $\beta_1$  constrained from OLS regression. The question is: what is the feasible range of values of  $E_0$  that could have given rise to the observed value  $P_0$ ? Here we call this the inverse prediction interval (IPI).

Quantifying the IPI is not trivial (e.g., Eisenhart, 1939; Hoadley, 1970, and see Osborne, 1991 for an excellent review). Frequentist inference, which is the primary statistical framework with which most scientists are familiar, is based on an assumption that the characteristics of the system have fixed values that can be constrained with the addition of data. As E is an independent variable, the uncertainty distribution parallel to this axis is not defined, so this statistical framework is not appropriate for estimating the IPI. This is the so-called "calibration problem" (Shukla, 1972).

There are, however, a number of alternative approaches to frequentist inference that can be used to estimate the uncertainty in the value of  $E_0$  for a given  $P_0$ . First, in the Bayesian statistical approach, the uncertainty distribution in  $E_0$  can be determined in an iterative fashion without formally rearranging Equation 1. Bayesian statistical approaches are well suited to the problem of inverse prediction, but numerical calculations can be time consuming, and the development of a model requires familiarity with Bayesian statistics. Second, non parametric approaches are based directly on observed distributions of the data. These approaches are powerful because they do not require assumptions about error structure within the data to be made, but they also require that the data comprehensively represent the system. Third, the "fiducial" approach, which was never widely adopted as a broad statistical framework, and is all but absent from the modern statistical literature, is nevertheless widely used in the specific example of uncertainty estimation in inverse-prediction problems (Williams, 1959). We refer the reader to (Draper & Smith, 1998) for a full description of this approach. Note that the "prediction interval" (PI), or root mean-squared error (RMSE) of an "E-on-P" OLS regression (i.e., with E on the y-axis and P on the x-axis) is not a valid estimate of uncertainty in  $E_0$ . This approach violates the assumptions of OLS regression, resulting in an incorrect slope and intercept, and an underestimate of the unexplained variance. Given the ubiquitous use of calibration-based estimates throughout the geosciences, and the inconsistent calculation of uncertainty, there is a pressing need for an uncomplicated, robust and generalizable way to estimate the IPI.

#### 3.2. Recommended Method for Simple Calculation of the IPI

We describe a simple and robust approach for estimating the IPI that can be straightforwardly applied to any calibration data set. In this approach we make the simplifying assumption that  $\beta_1$  and  $\beta_0$  are known with absolute certainty. Uncertainties in  $\beta_0$  and  $\beta_1$  are, of course, never zero, but this simplification is usually reasonable when the calibration data set shows a visually apparent correlation. If  $\epsilon$  is an approximately normal distribution, it will be best represented for a finite sample size by a Student's *t*-distribution. The value of  $E_0$  corresponding to a single value of  $P_0$ , to within an uncertainty  $\alpha$  is then:

$$E_0 = \hat{E}_0 \pm \frac{\hat{\sigma} t_{\alpha, n-2}}{\left| \hat{\beta}_1 \right|},\tag{3}$$

where  $\hat{E}_0$  is the classical estimator of E (Equation 2) and  $\hat{\sigma}$  is the observed standard deviation of the residuals from Equation 1.  $t_{\alpha,n-2}$  is the dimensionless critical value of the Student's *t*-distribution, where *n* is the number of calibration data points and a linear model has n - 2 degrees of freedom.  $\alpha$  is the level of confidence desired (i.e., for 95% confidence bounds,  $\alpha = (1-0.95)/2 = 0.025$ ). For  $n \to \infty$ ,  $t_{\alpha,n-2}$  approaches the critical value of the normal distribution ( $t_{0.025,\infty} = 1.96$ ). This estimate of uncertainty is shown in Figure 1 as "Simple IPI" and is graphically intuitive as rotating the standard deviation of the residuals from the *P* axis





**Figure 1.** *Proxy calibrations with associated inverse prediction intervals.* Linear regression analyses on an artificial data set of an environmental variable (*E*), and a measurable proxy variable (*P*). The artificial data set was generated with Gaussian noise in the *P* axis alone. The regression lines were calculated with a *P*-on-*E* ordinary least squares (OLS) linear regression. The confidence bands (CBs) of the regressions represent the 95% uncertainty in the model parameters intercept and slope. 95% inverse prediction intervals (IPIs) in *E* are given by four different approaches: 1. Simple IPI (this study: Section 3); 2. Nonparametric IPI (Text S1); 3. Fiducial IPI. 4. Bayesian IPI (strictly, credible interval; Text S2). The "true" parameters  $\beta_1$  (slope) and  $\sigma$  (noise), and the sample size, *n*, are prescribed for each condition. All approaches yield similar values of uncertainty in the estimated value of  $E_0$ , for a given value of  $P_0$  of 0.5, represented by horizontal lines at the bottom of each plot. (a) An ideal calibration consisting of 500 data points. (b) Calibration consisting of 30 data points picked randomly from (a). Note the line is less well constrained but the IPI is similar to (a). (c) Calibration with all parameters as (a), except a slope half as steep. Note the line is well constrained but the IPI is double that of (a). (d) Calibration with all parameters as (a), except a slope half as steep. Note the line is well constrained but the IPI is double that of (a).



onto the *E* axis (Figure S1). Equation 3 is equivalent to the "quick-and-dirty" estimate of the uncertainty in  $E_0$  given by Demidenko et al., which finds an estimate of  $\sigma$  for a Cauchy PDF in the limit of large *n* (Demidenko et al., 2013, Equation 16).

Some deductions can be made directly from Equation 3. The uncertainty in  $E_0$  is proportional to  $\sigma$ , and is independent of the intercept,  $\beta_0$ , but decreases with the magnitude of the slope,  $|\beta_1|$ . This inverse dependence leads to the correct units for  $\sigma / \beta_1$ , and inversion becomes impossible when the slope approaches zero. Even when parameters of the model ( $\beta_0$ ,  $\beta_1$  and  $\sigma$ ) are known exactly, uncertainty in  $E_0$  based on a single value of  $P_0$  is limited by the scatter of data about the regression line, and is not decreased by increasing the number of calibration data.

#### 3.3. Comparison of the Simple IPI Estimate With Alternative Approaches

Using an artificial calibration data set, we compare the Simple IPI to estimates from the well-established fiducial approach, a nonparametric approach, and a Bayesian approach. For the fiducial approach we used Equation 3.2.6 from Draper & Smith, 1998, which results in curved inverse prediction bands. For the non-parametric approach, we assumed a constant uncertainty across all E, and defined the IPI to lie within the 2.5th and 97.5th percentiles of the distribution of  $E_i - \hat{E}_i$  (i = 1, ..., n; see Text S1 for details). Bayesian calibration and prediction can be done: (a) in two steps (e.g., as in Tierney et al., 2019; Tierney & Tingley, 2014, 2018), whereby the posterior distributions of the parameters of the forward model are first established before they are used to generate posterior distributions of  $E_0$ ; or (b) in a single step (e.g., as in Gelman et al., 2004, 2020) whereby the posterior distributions of the parameters of the forward model and of the value of  $E_0$  are all generated within a single simulation. The Bayesian approach shown in Figure 1 is a two step approach (see Text S2 for details).

We furthermore evaluated all of these approaches against an "empirical" Monte Carlo (MC) simulation of the error distribution associated with inverse prediction. This involved repeatedly simulating the offset between a "known" value of *E*, and the value of *E* predicted from Equation 2 (see Text S3 and Figure S2 for details). The nonparametric approach produces a slightly asymmetric uncertainty range, and systematically underestimates the IPI at low *n* (Figure S3). The fiducial approach produces flared outward uncertainty range reflecting the effect of uncertainty in the slope, and systematically overestimates the IPI at low *n* (but is invalid at very low *n*; Figure S3). Estimates from the Bayesian and Simple IPI approaches lie closest to the simulated uncertainties. The Bayesian approach predicts slightly larger uncertainties at low *n*, and flares outwards very slightly due to uncertainties in the slope. All approaches converge on the same IPI as the MC simulation in the limit of large *n* (Figure S3). Further details of this simulation are given in the supporting information (Text S3). We also provide example *R* and *stan* (Stan Development Team, 2020) scripts for making these calculations (https://github.com/QGeoBio/IPI).

The calibration data set only needs to be large enough to define the three parameters of the model ( $\beta_0$ ,  $\beta_1$ , and  $\sigma$ ) with some confidence. Further increasing the size of the calibration data beyond this point does not significantly improve uncertainty in prediction (compare IPIs in Figures 1a and 1b and S3). For our artificial data sets, a "sufficiently large" data set to ignore parameter uncertainties is around 30–50 well-spaced data points (Figure S3), but the controls on this value of *n* are a subject of future research. For a single proxy measurement, the IPI is determined by the amount of unexplained variance in the calibration data set (more scatter = greater uncertainty; compare Figures 1a and 1c), and the slope of the regression (shallower slope = greater uncertainty; compare Figures 1a and 1d). Uncertainties in model parameters (which do change with increased sample size; compare CBs in Figures 1a and 1b) contribute only a small fraction of the total prediction uncertainty. From these comparisons, for calibration data sets exhibiting a visually reasonable correlation between *E* and *P*, the Simple IPI approach (Equation 3) is an excellent estimate of uncertainty.

#### 4. Example

Figure 2 shows a reconstruction of the surface ocean carbonate ion concentration ( $[CO_3^{2-}]$ ) across the termination of the last glacial cycle (Barker & Elderfield, 2002) using the foraminifera shell-weight proxy. This proxy is based on the size-normalized weight (SNW) of planktic foraminifera shells, which is shown to be





**Figure 2**. *Example:* Surface seawater carbonate ion concentration ( $[CO_3^{3-}]$  in  $\mu$ M) reconstructed using size-normalized foraminifera shell weights (SNW) over the last glacial cycle (Barker & Elderfield, 2002). The data shows a decrease in  $[CO_3^{3-}]$  from the last glacial maximum (~20 ka) to the mid-Holocene (~6 ka). Note that for the calibration we applied ordinary least squares (OLS) linear regression to the log transformed size-normalized weight (SNW) data whereas Barker and Elderfield (2002) fit an exponential curve to the SNW data. Inverse prediction intervals calculated were based on the calibration data set presented in the same study. See text for discussion and Figure 1 for details of each approach.

higher at higher  $[CO_3^{2^-}]$ . Based on this simple univariate core-top calibration, the 95% IPI is approximately  $\pm$  20  $\mu$ M (Figure 2a). The authors concluded that reconstructed  $[CO_3^{2^-}]$  decreased from the last glacial maximum (~20 ka) to the mid-Holocene (~6 ka) consistent with changes in carbonate chemistry expected from the timing and magnitude of CO<sub>2</sub> change observed in ice core records (Figure 2b). The inferred direction of the  $[CO_3^{2^-}]$  change is robust to the 95% level, but the magnitude of the change may be within a factor of 5 (max change =  $\Delta[CO_3^{2^-}] = 110 \ \mu$ M from 290 to 180  $\mu$ M; min change =  $\Delta[CO_3^{2^-}] = 20 \ \mu$ M from 240 to 220  $\mu$ M; see Figure S4).

If the following were true: (a) the nature of the samples in the calibration and the reconstruction were the same; (b) the signal was driven by  $[CO_3^{2^-}]$ ; and (c) the scatter in the calibration and the time series was all noise (i.e., random), the noise in the time series of SNW of planktic foraminifera would reflect the noise in the calibration (i.e.,  $\epsilon$ ). In this example, the coherency (autocorrelation) of the signal is very high, and the scatter in proxy data is relatively low. While this suggests that the signal in the proxy data is real (i.e., there was significant decrease in SNW of planktic foraminifera shells through this interval), there is a substantial uncertainty associated with how much of this signal was driven by  $[CO_3^{2^-}]$  (Figure 2b). The value of  $\epsilon$  derived from the core-top calibration is assumed to reflect natural variability, and its translation to an uncertainty in reconstructed  $[CO_3^{2^-}]$  implicitly includes the possibility that the observed trend in the SNW of planktic foraminifera shells could be driven by alternative additional influential variables (e.g., temperature [Qin et al., 2020] or nutrient concentration [Aldridge et al., 2012]).

# 5. Discussion

We restrict our discussion to the statistical perspective, with the following caveats: Calibrations, and the assumptions necessary for their application, must be representative throughout the geologic interval of interest. In other words, an assumption of uniformitarianism (sensu. Hutton [1785]) at the level of an observed relationship between *E* and *P* is the assumption that the model parameters,  $\beta_0$ ,  $\beta_1$  and  $\sigma$ , are constant through time. If the defined relationship is purely empirical, such that the mathematical form of the relationship is not mechanistically prescribed, and the coefficients of the relationship do not have



a physical meaning, it is difficult to evaluate the validity of this assumption, or justify extrapolation of the model beyond the calibration data. The integrity of the measured characteristic of the material on which the proxy is based must also be unchanged from the state in which it appears in the calibration, to the state in which it is used for a reconstruction, and not be significantly impacted by intervening processes (e.g., transport or diagenesis).

#### 5.1. Calibration Data: Experiments Versus Natural Samples

Calibrations based on experiments and calibrations based on natural samples provide different constraints on the model parameters. Constraining the coefficients of the linear model,  $\beta_0$  and  $\beta_1$ , is usually the primary goal of regression analysis. For this, laboratory experiments are arguably the most reliable approach because E can be varied in isolation and both E and P can usually be measured with low uncertainty. By contrast, in natural samples, environmental variables are often not measured directly, but instead are taken from global or regional databases that are binned in space and time, which introduces uncertainty in the value of E that corresponds to the measured value of P. For P-on-E OLS regression, uncertainty in E has the potential to result in an underestimate of the slope (regression dilution), and may necessitate an errors-in-variables alternative to regression such as reduced major axis regression (Draper & Smith, 1998), York regression or Bayesian methods. While experiments can be used to constrain predictor coefficients, in the real world P depends on various quantities that span the predictor parameter space. What if E-chosen because it is the environmental variable of interest—is not the variable exerting the strongest control on P in nature? The choice of cross-section through environmental parameter space represented by the experiments may lead to a significant interpretative bias during subsequent application if a calibration based on experiments is treated as directly representative of the natural world (e.g., Evans et al., 2018; Gray & Evans, 2019; Peterse et al., 2012).

Understanding the nature and magnitude of  $\epsilon$ , meanwhile, is an often overlooked goal of regression analysis. Consider *P* as a function P (*E*, *A*, *B*) where *A* are stochastic variables that are responsible for some noise even under ideal controlled laboratory conditions and *B* are stochastic variables that represent environmental factors other than *E* or *A*. *A* might reflect instrument precision or biological variability, while *B* reflects any environmental variable that can be held constant under laboratory conditions, but may vary in the natural world. In a linear regression model between *P* and *E*, *A* and *B* are treated as a combined noise term,  $\epsilon$ . Note that the Gauss-Markov assumptions for OLS regression require that the mean of the noise in the dependent variable must be everywhere zero—thus, values of the environmental variables reflected by *B* must themselves be uncorrelated with E. Variances are usually additive, so one might expect that  $\epsilon$  is larger in field samples compared to laboratory samples. Indeed, more unexplained variance in natural samples compared to laboratory experiments can be seen as an indication that another parameter not considered in laboratory studies is influential in the natural world. However, this is complicated by the potential for averaging (see below discussion).

Paleoenvironmental reconstructions always involve natural samples, so the value of  $\epsilon$  used to calculate the IPI includes both *A* and *B*— this is information that can only be generated with a calibration based on natural samples, as a controlled laboratory experiment only provides information on *A*. In theory, it would be possible to comprehensively explore proxy parameter space with experiments. In practice, this is very difficult to do. The dimensionality of the relevant parameter space is often unknown before it is explored, and exploration is limited by the constraints of time and cost of experiments. "Double calibrations", including both laboratory experiments and natural samples, may be a necessary compromise to isolating the effects of influential parameters, and estimating the fraction of real world variation that can be quantified.

#### 5.2. Noise Versus Systematic Uncertainty and Replicate Analyses

Under certain circumstances, the uncertainty in  $E_0$  can potentially be reduced by replicate measurements of  $P_0$ . However, whether this is possible depends on the nature of the unexplained variance in the calibration, and the nature of the noise in the paleoenvironmental samples. From the perspective of replication, the unexplained variance in the calibration can be thought of as a combination of: (a) noise, which is entirely stochastic, and which represents the range of measured values generated from repeated sampling of the "same" environment; and (b) systematic unexplained variance which is the result of variability in an influential environmental variable other than E. (a) would include all of the laboratory-based stochastic variables referred to as A above, but might also include some variables referred to above as B, including any natural events that are stochastic such as weather. (b) might include the effects of diagenetic status, or any environmental parameter that varies in a non-stochastic way in space and time, and between paleo-samples. The component of uncertainty due to unexplained variance of type (a) is reduced with replicate sampling, whereas that due to type (b) is usually not.

In principle, if the unexplained variance in the calibration was all noise ([a] above), and if the nature of the calibration data and the paleo-data were the same, the noise in a paleo-time series in *P* should reflect the noise in the calibration. In this ideal case, true replicates at each time point would reduce the uncertainty in the reconstruction. However, it is usually not possible to distinguish between (a) and (b), and similarities between the magnitude of the noise in the calibration and the paleo-data, are only circumstantial evidence of their common origin. Furthermore, there is ambiguity over the nature of a true replicate. Measurement replicates can account for instrumental error, and sample replicates can account for sediment heterogeneity, but are these true replicates of sampling the same environment? In reality, the definition of a true replicate depends on the proxy under consideration.

Further complications arise when the nature of the calibration samples and the target samples are different. For example, consider a hypothetical proxy based on material produced at the surface of a body of water. The calibration relies on sediment trap samples with seasonal changes in environment, while reconstructions are based on sediment samples. In this scenario, the analyzed material in both sediment trap samples and down core samples has been subjected to multiple influential environmental variables. However, each data point in the calibration represents a short interval of time (weeks to months), whereas a 1 cm depth interval of sediment would contain the integrated signal over years to tens of thousands of years, depending on the rate of sedimentation, and depth of bioturbation. If the reconstructed time series contains less noise than (unexplained variance in) the calibration, one possibility may be that the nature of the sediment samples smooths out noise or seasonal effects. Another explanation may be that the down-core signal is skewed toward one season because a larger fraction of material produced at one time of year is transported to depth and archived in the sediment. If the reduced scatter is due to a reduction in noise through temporal averaging, this might be reproduced in the calibration-and thus accounted for-through replicates, based on separate but closely spaced sediment traps (targeting the "same" environment). In the case of seasonal skew, it would be necessary to develop a mathematical model to relate the sub-annual surface signal to the export flux using additional information about the environment to determine the integrated contribution to the sediment throughout the year.

If a time series with substantially lower noise than the calibration is produced, either through the use of true paleo-replicates, moving average (MA) or local regression (LOESS or LOWESS) approaches, or as a result of smoothing through the nature of sediment accumulation, there can be some confidence that any signal in the data represents a real change. However, this does not help to constrain the nature of the change. Unless the impact of influential variables other than E can be quantified or discounted, the uncertainty in the reconstruction of a change in E must be large enough to include the possibility that it is a non-E environmental variable responsible for driving the observed change in P.

#### 5.3. Relative Environmental Change

Often, the goal of paleoclimate reconstruction is to determine the magnitudes of relative change of an environment over a period of time. A common misconception arises when relative change at the same core location is assessed. Intuitively, trends or anomalies in the reconstructions based on data from samples from the same core location should be associated with far smaller uncertainties than values compared between different sites. In this case relative change is often reported with only analytical uncertainties, or the uncertainty in the slope. To understand whether this is appropriate, consider an end-member scenario where the calibration data set consists of an infinite number of data points and there is zero uncertainty in the slope. It must be incorrect that zero uncertainty in the slope corresponds to zero uncertainty in the reconstructed value of *E*, because environments change in a multivariate fashion, and random events impact the value of *P*.



When the magnitude of the IPI is large relative to the magnitude of the signal in the proxy data, the reconstructed environmental change cannot be reported as robust. However, there is a subtle but fundamental distinction between asking (a) "what change in the environmental variable ( $\Delta E$  is reflected by the observed change in  $P(\Delta P)$ ?", compared with the much weaker question, (b) "what  $\Delta E$  could explain the observed  $\Delta P$ ?". The estimate is the same for both of these questions (calculated from Equation 2), but there is a large difference in the associated uncertainty. The uncertainty associated with (b) corresponds to the uncertainty in the slope of the calibration line ( $\beta_1$ ), while that associated with (a) has an uncertainty corresponding to the (much larger) IPI. This difference in uncertainties reflects the strength of the statement. When assessing the plausibility of a hypothesis, it may be appropriate to quote the uncertainty associated with (b). However, comparisons between different proxies, or between proxies and models, always requires that the uncertainty associated with (a) is used. We suggest that this distinction is made explicit whenever paleoclimate data are presented.

#### 5.4. Influence of Multiple Environmental Factors

Systematic uncertainty can be accounted for through the inclusion of additional explanatory variables in to a multiple regression model. This requires higher dimensional data sets and more complex models (e.g., Evans et al., 2016; Peterse et al., 2012; Tierney & Tingley, 2014). Note that the accurate attribution of the effect of each environmental variable on the value of *P* is limited if the explanatory variables are correlated (exhibit collinearity or temporal auto-correlation) but cannot be assumed to perfectly co-vary on all relevant spatial and temporal scales (Kutner et al., 2004). This is extremely common in natural environments. When multiple regression models (based on uncorrelated predictors) are used for reconstructions, explicit assumptions can be made about the likely ranges of these additional variables to constrain the parameter space.

A promising future direction to this approach could include the combination of multiple measurements with common explanatory variables into a multiple multivariate analysis. If *k* measurable characteristics of the same system  $(P^{(1)}, P^{(2)} \dots P^{(k)})$  are all functions of an equal number of environmental parameters  $(E^{(1)}, E^{(2)} \dots E^{(k)})$ , a model relating the control of  $E^{(1:k)}$  to  $P^{(1:k)}$  could be inverted to constrain  $E^{(1:k)}$  from measurements of  $P^{(1:k)}$  simultaneously. When multivariate data are considered together, there is potential to place tighter constraints on the system as a whole than is possible with a single explanatory variable alone.

#### 6. Summary and Outlook

In summary, any variance in the calibration that cannot be explained must be reflected in the uncertainty in the reconstruction. We provide a simple method to calculate this uncertainty (Equation 3). The focus must, therefore, be on accounting for as much of the variance in the calibration data as possible, either through appropriate replicate measurements, through additional explanatory variables, or through an explicit mechanism relating calibration and down-core data. The uncertainty associated with inverse prediction for a calibration with a single explanatory variable is not significantly improved with the addition of more data once the parameters of the model ( $\beta_0$ ,  $\beta_1$  and  $\sigma$ ) are reasonably well constrained. The controls on what constitutes a sufficiently large data set to ignore parameter uncertainties is a subject of future research. We, therefore, suggest that the emphasis should shift away from the generation of calibration data sets consisting of a very large number of data points comprising a single explanatory variable, and toward data sets that consist of fewer data points that comprise multiple explanatory variables.

In an ideal world, one would have the same level of noise in laboratory experiments, modern field data and paleo data. However, one would expect that the noise level increases from laboratory experiments to modern field data and further to paleo data because of additional influences of processes not represented in the laboratory or on short time scales. The differences in noise levels could provide clues about the nature of deviations from the ideal world. In reality, the noise level of paleo data might be even lower than those of the corresponding lab data because of smoothing due to averaging.

Paleoclimate proxies usually have calibrations that contain a relatively large amount of unexplained variance, which translates to large uncertainties in reconstructed values. We advocate for an approach where



the measurement data are presented as the direct observations, and the environmental reconstruction can, subsequently, be presented as a plausible interpretation. The uncertainty in the magnitude of environmental change that would *plausibly* explain the observation, corresponds to the uncertainty in the gradient ( $\beta_1$ ). Absolute reconstructions must carry the full associated uncertainty of the IPI. This distinction between assessing the plausibility of a hypothesis, and reconstructing absolute changes or states of environment, is subtle but of fundamental importance when comparing proxy records or integrating proxies with models.

#### **Data Availability Statement**

The data for the example in Figure 2 are available from Barker and Elderfield (2002); data were otherwise not used nor created for this research. *R* and *Stan* scripts for the calculations described in this manuscript are available on GitHub (https://github.com/QGeoBio/IPI), and an interactive web application is under development and will be available at: http://www.qgeobio.com/IPI.

#### References

Aldridge, D., Beer, C. J., & Purdie, D. A. (2012). Calcification in the planktonic foraminifera Globigerina bulloides linked to phosphate concentrations in surface waters of the North Atlantic Ocean. *Biogeosciences*, 9(5), 1725–1739. https://doi.org/10.5194/bg-9-1725-2012 Anand, P., Elderfield, H., & Conte, M. H. (2003). Calibration of Mg/Ca thermometry in planktonic foraminifera from a sediment trap time series. *Paleoceanography*, 18(2), 1050. https://doi.org/10.1029/2002pa000846

Barker, S., & Elderfield, H. (2002). Foraminiferal calcification response to glacial-interglacial changes in atmospheric CO<sub>2</sub>. Science, 297(5582), 833–836. https://doi.org/10.1126/science.1072815

Bemis, B. E., Spero, H. J., Bijma, J., & Lea, D. W. (1998). Reevaluation of the oxygen isotopic composition of planktonic foraminifera: Experimental results and revised paleotemperature equations. *Paleoceanography*, 13(2), 150–160. https://doi.org/10.1029/98PA00070 Casella, G., & Berger, R. L. (2002). *Statistical inference* (2nd ed.). Duxbury.

Dekens, P. S., Lea, D. W., Pak, D. K., & Spero, H. J. (2002). Core top calibration of Mg/Ca in tropical foraminifera: Refining paleotemperature estimation. *Geochemistry, Geophysics, Geosystems*, 3(4), 1–29. https://doi.org/10.1029/2001gc000200

Demidenko, E., Williams, B. B., Flood, A. B., & Swartz, H. M. (2013). Standard error of inverse prediction for dose-response relationship: Approximate and exact statistical inference. *Statistics in Medicine*, *32*(12), 2048–2061. https://doi.org/10.1002/sim.5668 Draper, N., & Smith, H. (1998). *Applied regression analysis*. Wiley.

Eisenhart, C. (1939). The interpretation of certain regression methods and their use in biological and industrial research. The Annals of Mathematical Statistics, 10(2), 162–186. https://doi.org/10.1214/aoms/1177732214

- Evans, D., Brierley, C., Raymo, M. E., Erez, J., & Müller, W. (2016). Planktic foraminifera shell chemistry response to seawater chemistry: Pliocene-Pleistocene seawater Mg/Ca, temperature and sea level change. *Earth and Planetary Science Letters*, 438, 139–148. https://doi. org/10.1016/j.epsl.2016.01.013
- Evans, D., Müller, W., & Erez, J. (2018). Assessing foraminifera biomineralisation models through trace element data of cultures under variable seawater chemistry. *Geochimica et Cosmochimica Acta*, 236, 198–217. https://doi.org/10.1016/j.gca.2018.02.048

Gafarian, A. V. (1964). Confidence bands in straight line regression. Journal of the American Statistical Association, 59(305), 182–213. https://doi.org/10.2307/228286910.1080/01621459.1964.10480711

Gelman, A., Carlin, J. B., Stern, H. S., & Rubin, D. B. (2020). *Bayesian data analysis* (3rd ed.). London: Chapman & Hall. http://www.stat. columbia.edu/~gelman/book/

Gelman, A., Chew, G. L., & Shnaidman, M. (2004). Bayesian analysis of serial dilution assays. *Biometrics*, 60(2), 407–417. https://doi.org/10.1111/j.0006-341X.2004.00185.x

Gray, W. R., & Evans, D. (2019). Nonthermal Influences on Mg/Ca in Planktonic Foraminifera: A Review of Culture Studies and Application to the Last Glacial Maximum. Paleoceanography and Paleoclimatology, 34(3), 306–315. https://doi.org/10.1029/2018PA003517

Gray, W. R., Weldeab, S., Lea, D. W., Rosenthal, Y., Gruber, N., Donner, B., & Fischer, G. (2018). The effects of temperature, salinity, and the carbonate system on Mg/Ca in Globigerinoides ruber (white): A global sediment trap calibration. *Earth and Planetary Science Letters*, 482, 607–620. https://doi.org/10.1016/j.epsl.2017.11.026

Hoadley, B. (1970). A Bayesian look at inverse linear regression. Journal of the American Statistical Association, 65(329), 356–369. https:// doi.org/10.1080/01621459.1970.10481085

Hutton, J. (1785). The system of the Earth, its duration and stability. As read by James Hutton at a meeting of the Royal Society of Edinburgh on 4 July 1785 printed and circulated privately.

Kim, J. H., Schouten, S., Hopmans, E. C., Donner, B., & Sinninghe Damsté, J. S. (2008). Global sediment core-top calibration of the TEX86 paleothermometer in the ocean. *Geochimica et Cosmochimica Acta*, 72(4), 1154–1173. https://doi.org/10.1016/j.gca.2007.12.010

Kluge, T., John, C. M., Jourdan, A. L., Davis, S., & Crawshaw, J. (2015). Laboratory calibration of the calcium carbonate clumped isotope thermometer in the 25–250°C temperature range. *Geochimica et Cosmochimica Acta*, 157, 213–227. https://doi.org/10.1016/j. gca.2015.02.028

Kutner, M., Neter, J., Nachtsheim, C., & Li, W. (2004). Applied linear statistical models (5th ed.). McGraw-Hill Education.

Osborne, C. (1991). Statistical calibration: A review. International Statistical Review/Revue Internationale de Statistique, 59(3), 309. https://doi.org/10.2307/1403690

Peterse, F., van der Meer, J., Schouten, S., Weijers, J. W., Fierer, N., Jackson, R. B., et al. (2012). Revised calibration of the MBT-CBT paleotemperature proxy based on branched tetraether membrane lipids in surface soils. *Geochimica et Cosmochimica Acta*, *96*, 215–229. https://doi.org/10.1016/j.gca.2012.08.011

Qin, B., Li, T., Xiong, Z., Algeo, T. J., & Jia, Q. (2020). Calcification of planktonic foraminifer Pulleniatina obliquiloculata controlled by seawater temperature rather than ocean acidification. *Global and Planetary Change*, 193, 103256. https://doi.org/10.1016/j. gloplacha.2020.103256

#### Acknowledgments

The authors would like to thank Peter Crockford, Sylvia Dee, Allison Jacobel, Wil Leavitt, and Max Llovd for comments on an earlier draft of the manuscript. Thanks to three anonymous reviewers whose comments greatly improved the manuscript. Thanks to Eli Tziperman for suggesting the nonparametric approach. H. McClelland acknowledges postdoctoral fellowships from the Weizmann Institute of Science, and the Israel Academy of Science and Humanities. I. Halevy acknowledges a Starting Grant from the European Research Council (OOID No. 755053). A. S. Bradley acknowledges funding from the US Department of Energy (DE-SC0014613) and the Simons Foundation (622740).



Scheffé, H. (1959). The analysis of variance. Wiley.

Shukla, G. K. (1972). On the problem of calibration. Technometrics, 14(3), 547-553. https://doi.org/10.1080/00401706.1972.10488944

Smith, R. J. (2009). Use and misuse of the reduced major axis for line-fitting. *American Journal of Physical Anthropology*, 140(3), 476–486. https://doi.org/10.1002/ajpa.21090

Stan Development Team. (2020). {RStan}: The {R} interface to {stan}. Retrieved from http://mc-stan.org/

Taylor, B. N., & Kuyatt, C. E. (1994). Guidelines for evaluating and expressing the uncertainty of NIST measurement results (pp. 1-20). National Institute of Standards and Technology.

Tierney, J. E., Malevich, S. B., Gray, W., Vetter, L., & Thirumalai, K. (2019). Bayesian calibration of the Mg/Ca paleothermometer in planktic foraminifera. *Paleoceanography and Paleoclimatology*, 34(12), 2005–2030. https://doi.org/10.1029/2019PA003744

Tierney, J. E., & Tingley, M. P. (2014). A Bayesian, spatially-varying calibration model for the TEX86 proxy. *Geochimica et Cosmochimica Acta*, 127, 83–106. https://doi.org/10.1016/j.gca.2013.11.026

Tierney, J. E., & Tingley, M. P. (2018). BAYSPLINE: A new calibration for the alkenone paleothermometer. Paleoceanography and Paleoclimatology, 33(3), 281–301. https://doi.org/10.1002/2017PA003201

Williams, E. (1959). Regression analysis. Wiley.